Author's response to reviews

Title: Bone mineral density, body mass index and cigarette smoking among Iranian women: Implications for prevention

Authors:

Azam Baheiraei (a.baheiraei@garvan.org.au)
Nicholas A Pocock (n.pocock@unsw.edu.au)
John A Eisman (j.eisman@garvan.org.au)
Nguyen D Nguyen (n.nguyen@garvan.org.au)
Tuan V Nguyen (t.nguyen@garvan.org.au)

Version: 4 Date: 2 May 2005

Author's response to reviews: see over
May 2 2004

The Editorial Team
BMC Musculoskeletal Disorders

Dear Editorial Team,

Ref: Ms: 7837286305676117
Re: Bone mineral density, body mass index and cigarette smoking among Iranian women: Implications for prevention

Thank you for your e-mail dated 26/4/2005 together with the Reviewers’ comments on the above manuscript. We have carefully considered the Reviewers’ comments and have addressed them in the following attachment. We have also modified parts of the manuscript to take into account the Reviewers’ concern. We hope that our response and the revised manuscript are satisfactory to you and suitable for publication in the Journal.

We look forward to hearing from you.

Yours sincerely,

Tuan V Nguyen
Response to Reviewer 1

Thank you for taking time to review our revised manuscript. We appreciate the Reviewer’s positive general comments.

Response to Reviewer 2

We appreciate the Reviewer for reviewing our revised manuscript. We initially could have misunderstood the Reviewer’s concern and have addressed differently. We think that we now understand the concern and have added some sentences in the Discussion section (page 11) to point out the potential weakness of the study as the Reviewer suggested.